Author’s response to reviews

Title: Serum and Synovial Fluid Cytokine Profiling in Hip Osteoarthritis: Distinct from Knee Osteoarthritis and Correlated with Pain

Authors:

Guomin Ren (guomin.ren@ucalgary.ca)
Ian Lutz (ial305@mail.usask.ca)
Pam Railton (prailton@shaw.ca)
J Preston Wiley (wiley@ucalgary.ca)
Jenelle McAllister (jmcallister@mtroyal.ca)
James Powell (jpowellemail@yahoo.ca)
Roman Krawetz (rkrawetz@ucalgary.ca)

Version: 3 Date: 24 Jan 2018

Author’s response to reviews:

1. Please change the 'Introduction' heading to 'Background'.

Completed

2. In the Authors' contributions section, we have noticed the initials 'SMI', however these initials do not correspond to anyone in the author list. Please clarify whether this is an author.

This must have been a typo as we dont have an SMI in our group. The SMI has been removed.

If you choose to change your author list you will need to fill out a change in authorship form and send it by email to the Editorial office to be approved by the Editor. The form can be found here: https://www.biomedcentral.com/getpublished/editorial-policies#authorship.

Please also include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

3. As a member of the editorial board (Roman Krawetz) of this journal, in order to ensure transparency, please declare this in the Competing Interests section of the Declarations.
4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Completed